Author's response to reviews

Title: Tracing the evolution of students' confidence in clinical and patient communication skills during a clinical internship: a multi-methods study

Authors:

Mark D Hecimovich (m.hecimovich@murdoch.edu.au)
Simone E Volet (s.volet@murdoch.edu.au)

Version: 2 Date: 16 October 2011

Author's response to reviews:

Dear Editor,

I had to revise the first sentence of the Background section due to poor structure.